Author's response to reviews

Title: Myositis ossificans occupying the thenar region: a case report

Authors:

Tsutomu Akahane (t.akahane@nagano-hosp.go.jp)
Naoya Mori (n.mori@nagano-hosp.go.jp)
Yukio Nakatsuchi (ynakatch@jeans.ocn.ne.jp)

Version: 3 Date: 23 March 2015

Author's response to reviews: see over
To The Journal of Medical Case Reports Editorial Team

Please find enclosed our revised manuscript entitled “Myositis ossificans occupying the thenar region.”, by Tsutomu Akahane, et al., for consideration of publication in the Journal of Medical Case Reports as a case report.

By re-submitting the manuscript to the Journal of Medical Case Reports, the authors understand that the material presented in this paper has not been published before nor has it been submitted for publication to another scientific journal or being considered for publication elsewhere. I attest that this work has been approved by all co-authors. The authors also understand that should the submitted material be accepted for publication in the journal, they will automatically transfer the copyright to the publisher.

I address the response to the concerns:

Which of the following following best describes what type of case report this is?:
- Unexpected or unusual presentations of a disease

Has the case been reported coherently?: Yes

Is the case report authentic?: Yes

Is the case report ethical?: Yes

Is there any missing information that you think must be added before publication?: No

Is this case worth reporting?: Yes

Is the case report persuasive?: Yes

Does the case report have explanatory value?: Yes

Does the case report have diagnostic value?: Yes

Will the case report make a difference to clinical practice?: Yes

Some typographical errors and grammatical errors:

We correct our revised edited by a native-English speaker with scientific expertise, Edanz group Japan.

Declaration of competing interests: I declare that I have no competing interests

I hope that the reviewing process finds the manuscript acceptable for publication in Journal of Medical Case Reports.

Sincerely Yours,

Tsutomu Akahane, MD
Dept. of Orthopedic Surgery
Shinshu-Ueda Medical Center
1-27-21 Midorigaoka, Ueda, Nagano, 386-8610, Japan
Tel.: +81-268221890. Fax: +81-268246603.
e-mail: t.akahane@nagano-hosp.go.jp